Author’s response to reviews

Title: An Investigation of Physical and Mental Health Consequences among Chinese Parents Who Lost Their Only Child

Authors:

qianlan Yin (yinquianlan@smmu.edu.cn)
weizhi Liu (liuweizhi@smmu.edu.cn)
Zhilei Shang (shangzhilei@163.com)
Na Zhou (nazhou@163.com)
liugang yu (liugangyu@smmu.edu.cn)

Version: 1 Date: 23 Feb 2017

Author’s response to reviews:

The article has been processed by the English-native writers, and made some revision.